Author's response to reviews

Title: Final 2 year results of the Vascular Imaging of acute Stroke for Identifying predictors of clinical Outcome and recurrent ischemic eveNts (VISION) study.

Authors:

Shelahg B Coutts (scoutts@ucalgary.ca)  
Michael D Hill (hillmd@ucalgary.ca)  
Misha Eliasziw (eliasziw@ucalgary.ca)  
Karyn Fischer (Karyn.Fischer@CalgaryHealthRegion.ca)  
Andrew M Demchuk (ademchuk@ucalgary.ca)  
for the VISION study group (scoutts@ucalgary.ca)

Version: 5 Date: 18 April 2011

Author's response to reviews: see over
We have made the suggested changes to the manuscript as follows:

Institutional addresses have been added for all authors. The VISION study group has been added as an extra author - this was problematic as there was not enough room. I have not removed the authors on the front page as this is important for the authors to be listed on the front. We would like the authors and the group listed as it is currently.

Background has been added to the introduction and conclusions at the end of the discussion.

References have been reformatted.

Track changes have been removed.